Author's response to reviews

Title: Pharmacokinetics of quinacrine in the treatment of prion disease

Authors:

Lotus Yung (lotusyung@yahoo.com)
Yong Huang (yhuang@itsa.ucsf.edu)
Pierre Lessard (lessard@itsa.ucsf.edu)
Giuseppe Legname (legname@itsa.ucsf.edu)
Emil T Lin (etlin@itsa.ucsf.edu)
Michael Baldwin (mikeab@itsa.ucsf.edu)
Stanley B Prusiner (stanley@itsa.ucsf.edu)
Chongsuk Ryou (cryou@itsa.ucsf.edu)
B. Joseph Guglielmo (bjg@itsa.ucsf.edu)

Version: 4 Date: 22 November 2004

Author's response to reviews:

1. Have further cropped the figures.
2. Have added "USA" to the Institute for Degenerative Diseases on the title page.